Daily Status Sunday 10-28-12

Weather

Partly cloudy, calm winds through the night, < 5 m/s,. Temperatures stayed around Oc.

Status
Qsoilc down again, Krypton @ 10m down, Vdsm6 down.
Looking at QC plots, the 2m TRH @9 has been consistently high compared to other stations. This agrees with the sonic temp, Tc, at 9

so | assume the TRH is good.

Update
Added conductive grease to cactus soil connectors. Qsoilc came back for a short period. The problem may be in the din connector on
the mote.

Removed Vdsm mote at 6 to test in the base. The unit is not putting out any messages.
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